Author’s response to reviews

Title: Modeling Dose-Response Relationships of the Effects of Fesoterodine in Patients With Overactive Bladder

Authors:

Linda Cardozo (lcardozo@compuserve.com)
Vik Khullar (vik.khullar@imperial.ac.uk)
Ahmed El-Tahtawy (ahmed.el-tahtawy@pfizer.com)
Zhonghong Guan (zhonghong.guan@pfizer.com)
Bimal Malhotra (bimal.k.malhotra@pfizer.com)
David Staskin (staskinatt@gmail.com)

Version: 4 Date: 12 August 2010

Author’s response to reviews: see over
Dear Rachel,

This letter accompanies the resubmission of the manuscript “Modeling dose-response relationships of the effects of fesoterodine in patients with overactive bladder” (MS 2061086907321615). The manuscript has been revised per the request of the BMC Series Editorial Production Team. A point-by-point description of the changes that were made is provided below.

Kind regards,

Colin Mitchell

On behalf of Linda Cardozo, MD
**Title page**

The title of the manuscript has been converted to sentence case.

The authors affiliations and email addresses are now provided in separate sections on the title page, with each author’s email address identified by their initials.

All underlining has been removed.

The word counts have been removed from the title page.

**Abstract**

The abstract now includes only 1 weblink.

**Methods**

Following the revisions to the Abstract, information has been added to the Methods section regarding trial registration at ClinicalTrials.gov for the phase III trials and the availability of the protocols of the phase II trials at ClinicalStudyResults.org.

The Appendix formerly identified as “Appendix” has been renamed as “Appendix II” (see below).

A typo was corrected (has changed to have).

**Discussion**

“…the modeling included data from patients with demography typically represented in OAB clinical trials (about 80% females and average age 65 yrs) predominantly female population…” was edited to read “…the modeling included data from patients with demography typically represented in OAB clinical trials (about 80% females and average age 65 yrs)…”

**Appendices**

Appendix I is now placed before the Acknowledgements section of the manuscript.
Appendix II has been renamed from Appendix. This appendix has been removed from the manuscript and uploaded as a separate file.

**References**

One of the references has been edited so that all authors' names are listed in place of the term et al.